Author's response to reviews

Title: Prevalence and Associated Risk Factors of Chronic Bronchitis in First Nations People

Authors:

Punam Pahwa (pup165@mail.usask.ca)
Chandima Karunanayake (chandima.karunanayake@usask.ca)
Donna Rennie (donna.rennie@usask.ca)
Joshua Lawson (josh.lawson@usask.ca)
Vivian Ramsden (viv.ramsden@usask.ca)
Kathleen McMullin (kathleen.mcmullin@usask.ca)
P. Jenny Gardipy (director@willowcreehealth.com)
Judy MacDonald (judithlynmacdonald@hotmail.com)
Sylvia Abonyi (sylvia.abonyi@usask.ca)
Jo-Ann Episkenew (firstnations.lung@usask.ca)
James Dosman (james.dosman@usask.ca)

Version: 1 Date: 01 Apr 2017

Author’s response to reviews:

Date: March 31, 2017

Editor

BMC Pulmonary Medicine

Re: Revised Manuscript submission

Manuscript# PULM-D-16-00386

‘Prevalence and Associated Risk Factors of Chronic Bronchitis in First Nations People’
Dear Editor;

Thank you for considering our manuscript entitled ‘Prevalence and Associated Risk Factors of Chronic Bronchitis in First Nations People’ for publication in your journal. We have considered and addressed the reviewers’ concerns. Please find attached:

• A point by point response to the reviewers’ comments, and
• the revised version of the manuscript.

Sincerely

Punam Pahwa PhD
Professor
Dept. of community Health and Epidemiology
Canadian center for Health and Safety in Agriculture
University of Saskatchewan
Phone: (306) 966 8300